Author’s response to reviews

**Title:** DNA methylation and histone modifications regulate SOX11 expression in lymphoid and solid cancer cells

**Authors:**

Lena Nordström (lena.nordstrom@immun.lth.se)
Elin Andersson (elinandersson74@gmail.com)
Venera Kuci (venera.kuci@immun.lth.se)
Elin Gustavsson (elin.gustavsson@roche.com)
Karolina Holm (karolina.holm@med.lu.se)
Markus Ringnér (markus.ringner@med.lu.se)
Per Guldberg (PerG@cancer.dk)
Sara Ek (sara.ek@immun.lth.se)

**Version:** 6  
**Date:** 3 March 2015

**Author’s response to reviews:**

Dear Editor,

Enclosed is our revised manuscript (7455806913592592), with the title “DNA methylation and histone modifications regulate SOX11 expression in lymphoid and solid cancer cells”.

The manuscript has been revised according to the comments. See point-by point answer below.

Reviewer’s comments: none remaining

Editorial comments:

1. **EDITORIAL REQUEST:**

Please specify whether informed consent was written informed consent and whether you obtained this consent from parents or guardians in the methods section of your manuscript.

This information has been added in the material and methods section. See line 315-316 and onwards.

Best Regards Sara Ek